Author's response to reviews

Title: Working practices and incomes of health workers: evidence from an evaluation of a delivery exemption scheme in Ghana

Authors:

Sophie Witter (sophiewitter@blueyonder.co.uk)
Anthony Kusi (tonykusi@gmail.com)
Moses Aikins (maikins@jsaconsultants.com)

Version: 2  Date: 19 October 2006

Author's response to reviews:

We have changed the title and made some minor corrections to figures in the text.